Dear Editor

The manuscript Fractures among patients with dizziness – a ten-year follow-up (BGTC-D-17-00084), has been resubmitted online.

The manuscript has been revised according to the reviewers suggestions. We wish to thank the reviewers for their time and their efforts to help us improve the manuscript. Please find a point-to-point list below with the revisions made.

The manuscript has not been published before and is not currently being considered for publication in any other journal.

There is no conflict of interest.

Corresponding author takes full responsibility for the contents of the paper.

We hope that you now find this manuscript acceptable for publication in BMC Geriatrics.

Eva Ekvall Hansson (corresponding author)
Associate professor in physiotherapy
Lund University
Department of Health Sciences
Health Science Centre
Box 157
SE-221 00, Lund, Sweden.
telephone:+46 46 2221986
e-mail:eva.ekvall-hansson@med.lu.se
Point-to-point list

1) P values are removed from the abstract

2) The spelling of deceased in table 1 is corrected

3) The sentence is changed and we hope that it is more clear now

4) Regarding the request for post-hoc power we find this a bit un-statistical, see http://daniellakens.blogspot.se/2014/12/observed-power-and-what-to-do-if-your.html

Or

http://core.ecu.edu/psyc/wuenschk/stathelp/Power-Retrospective.htm